Author's response to reviews

Title: A review of fatty acid profiles and antioxidant content in grass-fed and grain-fed beef

Authors:

Cynthia A Daley (cdaley@csuchico.edu)
Amber Abbott (aabbott1@exchange.csuchico.edu)
Patrick S Doyle (pdoyle@csuchico.edu)
Glenn A Nader (ganader@ucdavis.edu)
Stephanie Larson (slarson@ucdavis.edu)

Version: 5 Date: 9 March 2010

Author's response to reviews:

Dear Editorial Review

Re: 1233800112294314

On behalf of myself and my co-authors, I would like to re-submit the article entitled “A review of fatty acid profiles and antioxidant content in grass-fed and grain-fed beef” (previously entitled: “The added nutritional value of grass-fed beef”) with the following revisions.

Revisions include the following:

Corrected author initials on the title page
Added the competing interest section
Added the author contribution section
Added the acknowledgement section
Removed line numbering
Removed figures from the manuscript (uploaded as files)
Moved tables to follow the reference section
Removed any additional files
Corrected a couple of typographical errors

Sincerely,
Cynthia A.Daley